Author's response to reviews

Title: Meig's Syndrome presenting with Axillary Vein Thrombosis and Lymphadenopathy: A case report

Authors:

   Ridhima Iyer (ridhimapatel@gmail.com)
   Jason Chow (Jason.Chow@imperial.nhs.uk)
   Philip Savage (philip.savage@imperial.nhs.uk)

Version: 2 Date: 23 February 2013

Author's response to reviews: see over
Dear JMCR,

Meig's syndrome is rare and this particular case presented in a young patient in an atypical and clinically concerning manner.

The findings of venous thrombosis and regional lymphadenopathy made the index of suspicion of malignancy higher than usual and its description may help give reassurance in the management of any future similar cases.

We believe this is the first description of a case of Meig's syndrome presenting in this manner with venous thrombosis, pulmonary embolus and regional lymphadenopathy.

We hope this case report will be of use acute physicians, gynaecologists, radiologists and pathologists and are grateful for review in the JMCR

yours sincerely

Philip Savage
FRCP PhD
Consultant in Medical Oncology
Charing Cross Hospital
London